Author’s response to reviews

Title: Addressing safety risks in integrated care programs for older people living at home: a scoping review

Authors:

Manon Lette (m.lette@amsterdanumc.nl)
Eliva Ambugo (e.a.ambugo@medisin.uio.no)
Terje Hagen (t.p.hagen@medisin.uio.no)
Giel Nijpels (g.nijpels@amsterdanumc.nl)
Caroline Baan (Caroline.baan@rivm.nl)
Simone de Bruin (simone.de.bruin@rivm.nl)

Version: 2 Date: 14 Feb 2020

Author’s response to reviews:

Dear editor,

With this letter, we inform you that we have revised the manuscript titled "Addressing safety risks in integrated care programs for older people living at home: a scoping review." Please find below a point-by-point response to the issues raised. We hope to have addressed them satisfactorily and look forward to receiving your final decision.

Yours sincerely,

The authors

Response to Editor Comments:

1. Abbreviations: Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Authors: We added the abbreviations section, see Line 410. Furthermore, we checked whether these abbreviations were defined at first use, and added this definition in case it was missing.

2. Funding: In the Funding statement of the Declarations, please describe the role of the funding bodies in the design of the study, the collection, analysis, and interpretation of data, and in writing the manuscript.

Authors: The funder had no role in the design of the study, the collection, analysis, and interpretation of data, or preparation of the manuscript. We clarified this in Lines 427-428.
3. Full Declarations: Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

1. Ethics approval and consent to participate
2. Consent for publication
3. Availability of data and material
4. Competing interests
5. Funding
6. Authors' contributions
7. Acknowledgements

If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations. If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

Authors: We completed the Declarations section, please see Lines 416 - 439

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strike-throughs or text in different colours; this includes all supplementary material. All relevant tables/figures/additional files should also be clean versions. Please also ensure that all figures, tables and supplementary files are cited within the main body of text. Any items which are not cited may be deleted by our production department upon publication.

Authors: We uploaded a complete and clean file.